Author's response to reviews

Title: Better long-term survival in young and middle-aged women than in men after a first myocardial infarction between 1985 and 2006. An analysis of 8360 patients in the Northern Sweden MONICA Study

Authors:

Rose-Marie Isaksson (rose-marie.isaksson@nll.se)
Jan-Håkan Jansson (JanHakan.Jansson@vll.se)
Dan Lundblad (dan.lundblad@nll.se)
Ulf Näslund (ulf.naslund@vll.se)
Karin Zingmark (karin.zingmark@nll.se)
Mats Eliasson (mats.eliasson@nll.se)

Version: 3 Date: 3 January 2011

Author's response to reviews:

We have completed the following requests:

Authors’ contributions: Please indicate that all authors read and approved the final manuscript. DONE

Tables: We notice that table 1 has been created by using two separate tables. Please combine the two to create one whole table. Please also remove the visible vertical table lines from the table. DONE

Title: Please combine the subtitle and main title if you intend both lines of title to indicate the title. This will then mean it is identical to the title listed on the submission system. DONE

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. DONE